LEARNING THE HARD WAY: THE EFFECT OF VIOLENT CONFLICT ON STUDENT ACADEMIC ACHIEVEMENT

Tilman Brück

ISDC – International Security and Development Center, IGZ – Leibniz Institute of Vegetable and Ornamental Crops, Households in Conflict Network

Sami H. Miaari*

Tel-Aviv University and University of Oxford

Michele Di Maio

DSAE, University of Naples Parthenope

Abstract

We study the effect of the Israeli–Palestinian conflict on various education outcomes for Palestinian high school students in the West Bank during the Second Intifada (2000–2006). Exploiting withinschool variation in the number of conflict-related Palestinian fatalities during the academic year, we show that the conflict reduces the probability of passing the final exam, the total test score, and the probability of being admitted to university. The effect of conflict varies with the type and the timing of the violent events the student is exposed to and it is not significant for students in the upper tail of the test score distribution. We discuss various possible transmission mechanisms explaining our main result. Evidence suggests a role for both the conflict-induced deterioration of school infrastructures and the worsening in the student's psychological well-being due to direct exposure to violent events. (JEL: 120, O12, O15, F51)

The editor in charge of this paper was M. Daniele Paserman.

Acknowledgments: We thank the Editor (M. Daniele Paserman) and three anonymous referees for their constructive comments and suggestions. We thank the Palestinian Ministry of Education and Higher Education (MOEHE) for providing us with the data on exam results. We thank Applied Research Institute of Jerusalem for the superb help with the barriers' data. We thank Francesco Amodio, Muhammad Asali, Sonia Bhalotra, Massimiliano Cali, Luca de Benedictis, Francesco Drago, Belal Fallah, Nathan Fiala, Roberto Galbiati, Olaf de Groot, Kati Krähnert, Vincenzo Lombardo, Paolo Naticchioni, Roberto Nisticò, Giovanni Prarolo, Paola Salardi, Olga Shemyakina, Eik Swee, Chiara Toglia, Asaf Zussman, Noam Zussman, the participants at the XII Jan Tinbergen Conference, the 8th HiCN Conference, and seminar participants at GIGA (Hamburg) and at Humboldt University (Berlin). Sami Miaari thanks the European Union Seventh Framework Programme under grant agreement no. 218105 (EUSECON) for funding his stay at DIW Berlin while he was working on this project. Sami Miaari also acknowledges that this project has received funding from the Marie Sklodowska-Curie grant

agreement No. 752410. All errors are our own.

E-mail: brueck@isdc.org (Brück); michele.dimaio@uniparthenope.it (Di Maio); *Corresponding author: SamiMiaari@post.tau.ac.il (Miaari)

Journal of the European Economic Association 2019 17(5):1502–1537 DOI: 10.1093/jeea/jvy051 © The Authors 2019. Published by Oxford University Press on behalf of European Economic Association. This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs licence (http://creativecommons.org/licenses/by-nc-nd/4.0/), which permits non-commercial reproduction and distribution of the work, in any medium, provided the original work is not altered or transformed in any way, and that the work is properly cited. For commercial re-use, please e-mail: journals.permissions@oup.com

1. Introduction

Differences in academic achievement play a crucial role in explaining individual and cross-country differences in earnings and economic well-being (Card 1999; Hanushek and Woessmann 2011, Chap. 2). Individual academic achievement can be influenced by several factors such as personal, household, and school characteristics, as well as local and national socioeconomic conditions (Glewwe and Kremer 2006, Chap. 16). In this paper, we investigate one potentially important but little studied determinant of academic achievement, namely, living in and attending school in a locality affected by violent conflict.

Recent research on the microlevel effects of violent conflict provides evidence of the negative impact of the latter on education as measured by different outcomes, such as school enrolment, school attendance, and school attainment (UNESCO 2011a; Justino 2012, Chap. 13; Leon 2012; Buvinić et al. 2014). Yet, there are still significant knowledge gaps concerning both the effects of conflict on academic achievement and the mechanisms underlying such possible effects, in particular for high school students. Answering these questions is important because academic achievement is a predictor of future earnings, especially when admission to university is determined by high school final exam performance. Moreover, learning about the effects of conflict on high school students is crucial for a better understanding of the impact of violent conflict on the development prospects of an economy, given that these students will make up a large part of the country's high-skilled workforce in the future.

Despite the relevance of these questions, there are no studies on the effects of violent conflict on academic achievement of high school students. The main reason for that is a dearth of data. First, violent conflicts often occur in developing countries where collecting data on education and, in particular, on academic achievement is very difficult. The conflict situation usually exacerbates these difficulties. Second, the identification of the effect of conflict on academic achievement requires detailed data to measure conflict intensity. These are rarely available.

The characteristics of our two main data sets allow us to overcome the limitations of the literature. The first data set contains unique administrative data on individual test scores for the final high-school exam (Tawjihi General Examination) of the whole population of students enrolled in the Literary (Adabi) and the Scientific (Elmi) stream in high schools in the West Bank for all academic years of the Second Intifada (2000–2006). The Tawjihi General Examination is a crucial exam for young Palestinians. Completing high school education and getting a high test score at the final exam are expected to have a positive impact on the future earnings of the student, because these determine eligibility for admission to university in the West Bank and abroad, as well as for applying for well-paid public sector jobs. The second data set provides detailed information on all conflict-related Palestinian fatalities during the Second Intifada. These include the age, gender, and nationality of the victims, the date and locality of the events, as well as a brief description of the circumstances of the events. This comprehensive conflict data set allows us to construct an accurate measure of conflict

intensity at the locality level for each academic year, as well as for different time periods, types of victims and type of conflict events.

Taking advantage of these unique data, this paper provides the first empirical analysis of the effects of a violent conflict on academic achievement for high school students. Specifically, we study the effect of the Israeli–Palestinian conflict on individual results at the Tawjihi General Examination for the population of high school students in the West Bank during the Second Intifada, a period of intense violence between Israelis and Palestinians.¹ The West Bank during the Second Intifada provides a very suitable context for the analysis of the effects of violent conflict on academic achievement of high school students for various reasons. First, high school enrolment in the West Bank has always been high by international standards (UNESCO 2007). This provides us with a large population to analyze. Second, the functioning of the Palestinian education system was never interrupted during the Second Intifada, in contrast with many countries affected by extremely violent conflict. Third, the Second Intifada is a conflict characterized by meaningful temporal and geographical variation in violence intensity.

Our empirical analysis proceeds in two steps. First, we estimate the effect of conflict on individual academic achievement. We use spatial and time variation in conflict intensity—as measured by the locality-level number of all Palestinians killed by Israeli Defense Forces (IDF) during the academic year—to identify the effect of the Israeli–Palestinian conflict on individual exam results. By exploiting the within-school variation in the number of fatalities over time, we find that conflict reduces the probability of passing the exam, the total test score at the exam and the probability of achieving the minimum test score needed for admission to university. We provide evidence that these results are not due to sample selection, omitted variable bias, reverse causality, or migration choices. Moreover, we show that these results are robust to a time placebo test, to different samples and econometric models, and to the inclusion of controls for the level of internal mobility restrictions imposed by the IDF. We also document that the effect of conflict intensity is not significant for students in the upper tail of the test score distribution.

Second, we look for possible transmission mechanisms explaining why the conflict has a negative effect on academic achievement. As a first mechanism, we show that conflict negatively affects the quality of the learning environment at school. In particular, conflict increases student density in the classroom and this in turn correlates with a lower probability of passing the final exam. The second mechanism we explore is the worsening in student's psychological well-being due to exposure to conflictrelated violence. Direct exposure to conflict—as proxied by the per-capita number of fatalities in the locality—has a large negative impact on the probability of passing the exam, which effect is largest for conflict events occurring shortly before the exam date. Finally, we look at alternative possible mechanisms such as conflict-induced reduction

^{1.} The First Intifada—the name given to the first Palestinian uprising against the Israeli occupation of the Palestinian Territories—lasted from December 1987 until 1993.

in the number of school days, student and teacher absenteeism, reduction in quality of teaching and parental distress. Results indicate that conflict intensity seems to be associated with more school closures and more student and teacher absenteeism. Yet, due to severe data limitations on these variables, we cautiously interpret the overall empirical evidence on these mechanisms as being inconclusive.

Our paper builds upon the literature on the microeconomic effects of violent conflicts on education (see Blattman and Miguel 2010; Justino 2012, Chap. 13 (Verwimp et al. 2018) for reviews of the literature). The existing empirical evidence indicates that the effects of violent conflict on education are highly heterogeneous. The sign and magnitude of the effects depend, among other elements, on the characteristics of the country and of the violent events, on the age and gender of the student, as well as on the education outcome considered (see Grogger 1997; Blattman and Annan 2010; Chamarbagwala and Morán 2010; de Groot and Goksel 2011; Shemyakina 2011; Leon 2012; Rodriguez and Sanchez 2012; Di Maio and Nandi 2013; Valente 2013; Akbulut-Yuksel 2014; Justino et al. 2014; Swee 2015; Gershenson and Tekin 2018; Monterio and Rocha 2017; Michaelsen and Salardi 2018; Bertoni et al. forthcoming).

Our paper contributes to the existing literature in three ways. First, while the majority of previous studies look at the effect of conflict on the quantity of education (as measured by education enrolment, attendance or attainment), we estimate the impact of conflict on the quality of education (as measured by test score results at the school-leaving examination). Second, we provide novel evidence of the heterogeneous effect of conflict at the individual level. Our unique data allow us to explore aspects that the previous studies, using school-level observations as the unit of analysis, could not consider. In particular, we document that conflict has a heterogeneous effect along the test score distribution and on the type and timing of the conflict events the student is exposed to. Finally, we are able to explore various possible mechanisms. In particular, the finding that conflict reduces academic achievement, because it worsens the learning and teaching environment conditions by damaging school and classroom infrastructures, is new to the literature, and has important policy implications, especially in post-conflict situations. Moreover, to the best of our knowledge, our analysis of the psychological mechanism, and the suggestive evidence we provide supporting it, is also new to the literature.

The paper proceeds as follows. The next section provides some background on the Israeli–Palestinian conflict and the Second Intifada, as well as on the main characteristics of the Palestinian education system and of the Tawjihi General Examination. Section 3 describes the data. Section 4 discusses the econometric model and the identification strategy. Section 5 presents the regression analysis and the results. Section 6 discusses the possible transmission mechanisms. Section 7 concludes.

2. Background

As a result of the Six-Day War in 1967, the West Bank and the Gaza Strip were occupied by Israel. In 1993, the Oslo Accords created the Palestinian National Authority giving

Palestinians control over some administrative functions (e.g., education, health, and taxation) in both the West Bank and the Gaza Strip (henceforth the Occupied Palestinian Territories (OPT)) for the first time. Israel retained control over the strategic issues of security, foreign trade, and border controls.

The Second Intifada (also known as the Al-Aqsa Intifada) started in September 2000, ending the relatively peaceful period that followed the Oslo Accords. The Second Intifada was a violent revolt against the Israeli occupation of the OPT characterized by frequent clashes between Palestinians and the IDF.² During this period, numerous violent acts were perpetrated by both the Palestinians and the IDF, including the killing of civilians and Palestinian militants in the OPT, Palestinian suicide attacks in Israel, assassinations of Palestinians leaders in the OPT, and the demolition of Palestinian houses by the IDF.

In terms of number of fatalities, the conflict was highly asymmetric. Between 2000 and 2006, Palestinians killed 703 Israeli civilians and 316 IDF soldiers, whereas the IDF killed more than four thousand Palestinians (B'Tselem 2007). Although the intensity of violence varied over time and space, the conflict persisted throughout the entire period. While there is no official end date for the Second Intifada, violence decreased substantially after 2006. The results of the 2006 Palestinian legislative elections caused a de facto division of the OPT into a Fatah-controlled West Bank and a Hamas-controlled Gaza Strip. Starting from January 2007, Israel imposed a complete blockade on the Gaza Strip as a sanction against Hamas. After that, the West Bank and the Gaza Strip started to diverge in economic and political terms (Etkes and Zimring 2015).

Since the Oslo Accords, the education system in the OPT has been managed by the Palestinian Ministry of Education and Higher Education (MOEHE). The Palestinian education system includes government schools, private schools, and UNRWA (United Nations Relief and Works Agency for Palestine Refugees in the Near East) schools. In the West Bank, education is compulsory for ten years starting from age six.³

The secondary education cycle consists of either a vocational or an academic track and includes grades 11 and 12. High school students in the academic track can choose between two streams: Adabi (arts) and Elmi (scientific).⁴ Government schools and private schools offer high school grades for both streams whereas UNRWA schools do not provide services beyond grade 10. There are no differences in the requirements for entering the two streams. The geographical distribution of the schools

^{2.} For a description of the different periods of violence during the Second Intifada, see Jaeger and Paserman (2008).

^{3.} The education system is divided into three levels: elementary school (grades 1–6), middle school (grades 7–10), and high school (grades 11 and 12). For a detailed analysis of the Palestinian education system, see Nicolai (2007).

^{4.} One difference between the two streams of study is that graduates holding the Elmi Tawjihi Certificate may apply for all majors at the university, whereas graduates with the Adabi Tawjihi Certificate are limited in their choice to majors taught in the faculties of arts, economics, business administration, and law.

offering each stream is homogenous (World Bank 2011) and the fee to be paid is the same.⁵

At the end of these two years, all students take a final exam called the Tawjihi General Examination. The exam takes place on the same day (at the end of the academic year around the end of June) in all schools in the West Bank under exactly the same conditions. An external commission nominated by the MOEHE grades the exam. The exam is different for the Adabi and Elmi streams in terms of both subjects and for the weight given to each subject in the total test score (see Online Appendix B for details). The maximum test score in both streams is 1000. To pass the final exam, the student has to get more than 50% of the score in all subjects included in the test. Students who do not pass the exam can retake it the next year.

The Tawjihi Certificate is equivalent to a high school completion certificate and is required for students who wish to pursue undergraduate studies. The exam test score determines the likelihood of being accepted by a university in the OPT or abroad and which majors the student can enter. In the West Bank, the minimum score at the Tawjihi General Examination required for admission to university is 650, although some specific majors have higher minimum test scores for admission. For instance, the minimum test score for admission to engineering is 850. Similarly, the minimum test score required for the admission as a foreign student to most universities in other countries in the Middle East is 850.

Education is given much importance in the West Bank (Nicolai 2007). The number of enrolled students in all grades has been constantly increasing in the OPT, at least since the creation of the national curriculum following the 1994 Oslo Accords. This resulted in enrolment and attendance rates, and number of years of schooling, that are high by regional standards (UNESCO 2007). According to data by the Palestinian Central Bureau of Statistics (PCBS various years), the percentage of students enrolled in high school during the Second Intifada was between 72% and 83%,⁶ a very high level for a developing country.⁷ The high value placed on schooling also extends to the education of girls. In fact, the female participation rate in secondary education is slightly higher than male one (Di Maio and Nisticò 2018).

High school education has several perceived benefits in Palestinian society (World Bank 2006). Completing high school education and receiving a high test score at the final exam are important for at least three reasons. First, completing high school is the necessary step to access university. This is an attractive choice since university

1507

^{5.} Data from the preconflict census indicate that more than 86% of the West Bank population lives in a locality with at least one high school (PCBS 1997). In fact, according to the World Bank (2007b), access to high school education is highly equitable with respect to location.

^{6.} The upward trend in enrolment that characterizes the period 2000–2006 mirrors the increase in population growth in the decade before the Second Intifada (Nicolai 2007; World Bank 2007b; UNICEF 2010). World Bank (2002) reports that the average population growth rate before the Second Intifada was about 4% per year, one of the highest in the world, with about half of the population being under 15 years old. We discuss this issue in detail in Section 4.

^{7.} As for comparison, the percentage of secondary enrolment in 2006 for Egypt is 81 and for Lebanon 77, whereas the average for Arab countries is 67 (UNESCO 2007).

	Observations	Mean	Std. Dev.	Min	Max
Pass exam	138,135	0.74	0.4	0	1
Total test score	138,135	640.51	213.2	0	997
Age	138,135	18.32	0.6	17	28
Male	138,135	0.45	0.9	0	1
Adabi	138,135	0.73	0.4	0	1
Christian	138,135	0.02	0.1	0	1
Number of Fatalities (locality-year level)	1398	7.71	17.1	0	105

TABLE 1. Descriptive statistics: Main variables.

Notes: *Pass exam* is a dummy that takes value 1 if the student has passed the final exam (i.e., the student gets more than 50% of the scores in each subject at the exam) and 0 otherwise. *Total test score* is the standardized total test score at the exam. *Number of fatalities* is the locality-level number of conflict-related Palestinian fatalities caused by the IDF during the academic year.

education is practically free of charge (EACEA 2012) and private rates of return for university graduates during the Second Intifada were higher than for any other education level (Al-Kafri 2004; World Bank 2006; Tansel and Daoud 2011). Second, higher education is a requirement to access most of the public sector jobs in the OPT. Wages in the public sector are higher than in the private sector and public sector workers receive important work-related benefits and privileges (ETF 2014; Miaari 2018). Moreover, while the private sector has been dramatically affected by the conflict (Amodio and Di Maio forthcoming), this was much less so for the employment and wages in the public sector (Miaari 2009; Cali and Miaari 2018). Third, obtaining the Tawjihi Certificate with a high total test score is a necessary condition to access any university abroad. This opportunity, although being an important career opportunity for many students, is particularly attractive to those living in a conflictaffected country because it would allow moving to a safer environment. In summary, students completing high school education and receiving high test scores are expected to have higher future earnings.

3. Data

Our analysis combines four data sets. The first data set includes unique information on individual test scores in all subjects at the high-school final exam (Tawjihi General Examination) for the whole population of Palestinian students in the West Bank for all the academic years 2000–2006. These data are provided by MOEHE. There are 138,135 students included in our analysis. Our main variable of interest is *Pass exam*, a dummy that takes value 1 if the student passes the final exam and 0 otherwise. As shown in Table 1, 74% of the students have passed the final exam. The average total test score at the exam is 640. The data set also includes information on age, gender, religion, and stream of study (Adabi or Elmi) of the student as well as the identification

number and locality⁸ of the school the student is enrolled in. Summary statistics are reported in Table 1. The average age is 18.3 with 95% of the students being between 17 and 19 years old. The majority of the student population is female (55%). More than 97% of the students are Muslims and 2% are Christians. Students enrolled in the Adabi stream comprise 73% of the total.

The second data set—also provided by the MOEHE—includes information on the characteristics of the 555 high schools in the West Bank, including the locality of the school, the type of stream offered (Adabi/Elmi) and the supervising authority (government/private). As for the stream of study, 274 schools offer both the Adabi and Elmi streams, 260 only the Adabi stream, and 21 only the Elmi one. 9% of schools are private. The data set also reports the following characteristics for all schools: *Student/Teacher Ratio, Average Size of the Classrooms, Average Density in the Classroom, Average (years of) Teaching Experience, Percentage of Christians in the School*, and *Percentage of Dropout* at the school level.⁹ Table A.1 in Online Appendix A reports the descriptive statistics for these variables for the whole Second Intifada period (2000–2006). We merge this data set on school characteristics with the data set containing the individual test scores by the identification number of the school.

The third data set contains information on all Palestinian fatalities during the Second Intifada. Data on fatalities related to the Israeli–Palestinian conflict are collected and distributed by the Israeli NGO B'Tselem (Israeli Information Center for Human Rights in the Occupied Territories).¹⁰ For each fatality, the B'Tselem data set reports name, age, and gender of the victim, location of the event, and a description of the circumstances of the event. The information on each fatality is based on B'Tselem's investigation into the circumstances of each death occurred during a conflict event.¹¹ B'Tselem crosschecks its information with official IDF announcements, information gathered by Palestinian and international human rights organizations and media reports. B'Tselem data on fatalities are considered to be accurate and reliable by both Israelis and Palestinians (Mansour and Rees 2012) and have previously been used by other scholars studying the Israeli–Palestinian conflict (e.g., Jaeger and Paserman 2008; Jaeger et al. 2012).

Using the available information for each fatality, we construct our main variable to measure conflict intensity as the total number of all Palestinians killed by the IDF in each locality during each academic year.¹² We merge our conflict intensity variable

^{8.} The locality is the smallest administrative unit used by the PCBS. There are 610 localities in the West Bank.

^{9.} See Section 6.1 for the definitions of these variables.

^{10.} Data on fatalities for the Israeli–Palestinian conflict are available at this link: http://www.btselem.org/statistics.

^{11.} See http://www.btselem.org/statistics/casualties_clarifications.

^{12.} Previous studies on the Israeli–Palestinian conflict used a variety of variables to proxy conflict intensity: number of closure days, number of Palestinian suicide bombings in Israel, and so forth. With respect to those, the locality-level number of Palestinian fatalities provides a significantly more accurate measure of individual exposure to conflict intensity (see Brück et al. 2016).

with the other two data sets using the information on the locality of the event. Table A.2 in Online Appendix A reports the descriptive statistics for Palestinian fatalities between the beginning of the Second Intifada in September 2000 and June 2006.¹³ After the rapid increase in the number of fatalities after the start of the Second Intifada, the conflict reaches a peak in 2002 and then decreases in intensity until 2006. The total number of Palestinian fatalities during the period under consideration is 1495. As reported in Table 1, the yearly average number at the locality level is 7.7, its standard deviation is 17.2 and the maximum is 105. This indicates a large variation in students' exposure to conflict across localities and within localities over time.

The fourth data source is the Palestinian Labor Forces Surveys (PLFS) for the period 2000–2006, as provided by the PCBS. The PLFS is a representative household survey of Palestinians living in the OPT. The PLFS is a rotating quarterly panel where households are surveyed four times over six quarters. They are surveyed for two consecutive quarters, not surveyed in the next two quarters, and then surveyed again for two consecutive quarters. The survey collects quarterly data on employment and socioeconomic characteristics of individual household members aged 15 or older. The survey also reports school attendance status and the number of completed years of education for each individual. We use PLFS data to discuss the possible issue of sample selection in our main data set and to compute various economic variables (e.g., district-level wage, low-skilled wage, and unemployment rate) to explore the relationship between conflict intensity and local-level economic conditions.

4. Econometric Model and Identification Strategy

The sign of the relationship between conflict intensity and student academic achievement is not obvious. On the one hand, conflict may negatively affect the performance of the student at the final exam. For instance, the conflict may make the learning process more difficult (by deteriorating the quality of the school environment or by causing the student to suffer psychological distress), or it could reduce the incentives to attend school (by reducing the value of education or increasing the value of the student's outside option), thus lowering the probability of passing the exam. On the other hand, the conflict may provide a strong incentive to increase the learning effort and to perform well at the final exam, because this could offer the opportunity to access well-paid jobs after school completion or to escape the conflict situation by studying abroad. Moreover, it is also possible that the effect of the conflict is heterogeneous across students and that the sign and magnitude of its impact depend on individual characteristics such as academic ability or gender. Only a rigorous analysis can discriminate between these theoretical predictions.

^{13.} The Second Intifada began on September 29, 2000 but there is no official end to it. Although some indicate 2005, the violence continued on both sides throughout 2006. In fact, the number of Palestinian fatalities in 2006 was three times that of 2005 (B'Tselem 2007). Since our results are not affected by this choice (results available upon request from the authors), we also include year 2006 in our analysis.

Using repeated cross sections of high-school final exam results from 2000 to 2006 in the West Bank, our identification strategy exploits the within-school variation over time in the locality-level number of Palestinian fatalities occurring during the academic year in which the student takes the exam. We estimate the effect of the conflict on individual academic achievement as follows:

$$Result_{islt} = a + b * Number of fatalities_{lt} + X'_{it} * c + Z'_{lt} * d + \mu_c + \gamma_s + \partial_t + u_{islt}, \qquad (1)$$

where *Result_{islt}* is the education outcome of interest (passing the exam, total test score, math test score, Arabic test score, minimum score for the admission to university or minimum score for the admission to university abroad) for individual *i* enrolled in high school s in locality l in academic year t. Number of fatalities_l, is the number of all Palestinian fatalities in school locality l caused by the IDF during the academic year t divided by 1000. X_{it} is a vector of individual student characteristics, such as age, gender, stream of study, and religion. Z'_{lt} includes a set of local-level economic variables to control for time-varying local economic conditions and the value of the student's outside options. Because the age at which the final exam is taken varies significantly across individuals, we include birth-cohort fixed effects (μ_c) to control for any shock common to all students born in the same year. We also include schoollevel fixed effects (γ_{c}) and academic-year fixed effects (∂_{t}). School-level fixed effects control for unobservable time-invariant differences across schools that may influence student performance at the final exam. Moreover, they allow us to control for timeinvariant characteristics of the school locality that may influence both the level of conflict intensity and student's performance. The academic-year fixed effects account for differences in the content and difficulty of the final exam in different years. They also control for the influence on student exam results of all the time-varying macroeconomic conditions such as labor market conditions and national education policies. Finally, u_{islt} is the error term. All regressions are estimated using OLS with robust standard errors clustered at the locality level, the level at which we measure conflict intensity.

5. Results

5.1. The Impact of Violent Conflict on the Final Exam Outcome

In our baseline specification of model (1), we use as dependent variable *Pass exam*, a dummy that takes value 1 if the student passes the final exam and 0 otherwise. Table 2 reports the within-school estimates of the effect of conflict intensity on the individual probability of passing the exam. Column (1) shows the result for a specification that includes only birth-cohort, school-, and year-fixed effects. The estimated coefficient for *Number of fatalities* captures the difference in percentage points in the probability of passing the final exam for students enrolled in the same school in different years that is only due to the difference in the number of Palestinian fatalities occurring in the

		Pa	ss exam	
		Full sample		First-time takers
	(1)	(2)	(3)	(4)
Number of fatalities	-0.43***	-0.49***	-0.52***	-0.45***
	(-0.14)	(-0.15)	(-0.15)	(-0.14)
Male	. ,	-0.06***	-0.06***	-0.07***
		(-0.01)	(-0.01)	(-0.01)
Christian		-0.01	-0.01	-0.01
		(-0.02)	(-0.02)	(-0.02)
Adabi		-0.19***	-0.19***	-0.18***
		(-0.01)	(-0.01)	(-0.01)
District-level economic controls	No	No	Yes	Yes
Birth-cohort fixed effects	Yes	Yes	Yes	Yes
School-fixed effects	Yes	Yes	Yes	Yes
Academic-year fixed effects	Yes	Yes	Yes	Yes
Number of observations	138,135	138,135	138,135	103,756
Mean of the dependent variable	0.74	0.74	0.74	0.78

TABLE 2. Effect of conflict intensity on the probability of passing the high-school final exam for Palestinian students in the West Bank during the Second Intifada.

Notes: OLS regression results for equation (1). For all regressions, the dependent variable is *Pass exam*, a dummy that takes value 1 if the student has passed the final exam and 0 otherwise. *Number of fatalities* is the locality-level number of conflict-related Palestinian fatalities caused by the IDF during the academic year divided 1000. *Male* is a dummy that takes value 1 if the student is male and 0 otherwise. *Christian* is a dummy that takes value 1 if the student is christian and 0 otherwise. *Adabi* is a dummy that takes value 1 if the student is enrolled in the Adabi stream and 0 if enrolled in the Elmi stream. *District-level economic controls* include district-level average wage, district-level low-skilled wage and district-level unemployment rate. Column (4) includes only *first-time exam takers*, that is, students who are 17 or 18 at the exam. In each regression, a constant is included but not shown. Robust standard errors in parenthesis are clustered at the locality level. ***Significant at 1%.

school locality during the academic year. The results shown in the table indicate that the impact of conflict on the probability of passing the exam is negative and highly significant.

The negative effect of conflict on final exam result is robust to the inclusion of several control variables. In column (2), we add a set of student characteristics.¹⁴ Results indicate that the probability of passing the exam is lower for male students and for those enrolled in the Adami stream. In column (3), we include a set of district-level economic variables, such as the average wage, low-skilled wage and unemployment rate, to control for time-varying local economic conditions and the student's outside options. The magnitude and significance of the effect of conflict on the probability of passing the exam remains highly significant and slightly increases.

^{14.} The MOEHE data set does not include information on the socioeconomic condition of students. Although this is a limitation of the data set, we argue that since our analysis focuses on students enrolled in nonmandatory high school, this information is somehow less relevant than for other studies. In fact, the variance in household income for these students is likely to be much smaller than is the case for, for example, primary school students.

Computing the average magnitude of the within-school estimates of the effect of conflict, we find that one standard deviation increase in the number of fatalities occurring during the academic year at the locality level reduces the probability of passing the exam by 1 percentage point $[(-0.52/1000) \times (17.2) = -0.01]$. As the mean of *Pass exam* is 0.74, this implies that a one standard deviation increase in the *Number of fatalities* reduces the probability of passing the exam by 1.2% of the mean.¹⁵

Finally, in column (4), we report the results when we restrict the sample to students taking the exam for the first time.¹⁶ The impact of conflict on the probability of passing the exam remains negative and highly significant whereas its magnitude is slightly reduced.

5.1.1. Threats to Identification. Our identifying strategy is valid under the assumption that, conditional on school- and time-fixed effects and controlling for observable individual characteristics, the within-school variation in conflict intensity over time is orthogonal to any other determinant of student results at the exam. There are four possible reasons why this assumption may not hold.

First, it is possible that our analysis suffers from a sample selection bias due to conflict affecting the characteristics of the pool of students taking the final exam. There are at least two aspects to consider. The conflict could affect the characteristics of students who enroll in high school. Furthermore, conflict may select the students who are actually taking the exam at the end of the academic year. As for the first aspect, the concern is that low-ability students who in the absence of the conflict would have abandoned school after the compulsory years may instead decide to continue schooling because the conflict reduces out-of-school occupational opportunities. In this case, the finding of a lower academic performance in localities with higher conflict intensity would just be the result of the conflict lowering the average ability of students enrolling in high school in those localities.¹⁷

^{15.} To the best of our knowledge, the only other study looking at the effect of conflict on high school completion is Swee (2015). In the case of the Bosnia–Herzegovina War, a one standard deviation increase in the casualty rate reduces the probability of completing high school by 3.8 percentage points. This larger effect is due to the fact that those who do not enroll or dropout during the academic year are also included among noncompleting individuals.

^{16.} First-time takers should be 18 years old at the exam. In our sample, 3% of students are 17 years old at the exam. We define first-time exam takers as the students that are 17 or 18 years old at the day of the exam. Data show that, although 25% of students take the exam at least one year later than expected, only 4% of students take the exam more than two years later than expected.

^{17.} Another reason why conflict may increase the number of low-ability students in high school is that it may induce parents to keep them in school as a way of protecting them from violence. Although this may be a common strategy in other conflict-affected environments, this is not the case for the West Bank. In fact, the conflict situation made travelling between home and school risky and difficult. For example, MOEHE reports 132 students killed and 2,500 injured while traveling to or from school in 2003. Schools were also not safe places: during the Second Intifada, schools suffered attacks and were occupied by the military. Akesson (2014) reports that students felt unsafe at school and parents were often reluctant to send children to school because commuting was considered dangerous.

Unfortunately, the MOEHE data set includes information only on students who actually took the exam. This implies that we cannot compare the ability level of enrolled individuals versus those who are not. Thus, to explore how conflict affects the probability of continuing to high school, we use PLFS data to estimate the determinants of the transition from compulsory to noncompulsory education. We take advantage of the fact that the PLFS is a rotating quarterly panel and define a variable capturing the change in schooling status from one year to the other. To identify those individuals who are in their last year of compulsory education, we focus on students aged 15 and 16 with at least 9 years of completed education at the first interview. We measure conflict intensity as the cumulative number of fatalities occurred during the twelve months before the last interview in the district of residence of the student.

Results reported in Table A.3 of Online Appendix A, column (1) indicate that the probability of continuing to high school after compulsory education is uncorrelated with conflict intensity, whereas it is positively correlated with the level of household head's education. In column (2), we include the interactions of conflict intensity with the level of household head's education level and with the district-level economic conditions, our proxies for the value of the student's outside options. Results are unchanged with respect to column (1) and indicate none of the interaction terms is significant. We interpret this evidence as indicating that students continuing to high school are those from higher-educated and wealthier households. Moreover, these results do not support the possibility that conflict induces lower-ability or disadvantaged students to stay in school by changing the out-of-school occupational opportunities.¹⁸

As noted previously, conflict may also select the sample of students who actually take the exam at the end of the academic year. In this case, the concern is that, if conflict makes higher-ability students to drop out during the academic year, our finding of a negative effect of conflict on the probability of passing the exam could just be the consequence of the lower average ability of the students taking the exam.

To test for this possibility, we perform two exercises. First, we estimate the determinants of school attendance for students in high school using the PLFS data set. To this end, we consider individuals aged 16, 17, and 18 living in the West Bank who have completed 10 years of education and are attending school. Results shown in Table A.4 of Online Appendix A indicate that high school attendance is uncorrelated with conflict intensity, whereas it is positively correlated with the household head having secondary or higher education.

Second, we look at the effect of conflict on the number of students actually taking the final exam at the end of grade 12 using the information included in the MOEHE data set. To test for this, we use the school-level *Number of exam takers* and *Percentage of dropouts*, that is, percentage of individuals who are enrolled but abandon school during the academic year. We regress these measures on the locality number of fatalities controlling for local economic conditions and school- and academic-year fixed effects.

^{18.} These results are consistent with the evidence reported in UNICEF (2010) that the primary reasons for dropping out after compulsory schooling years are the need to work to help the family and for early marriage.

Results reported in Table A.5 of Online Appendix A show that the within-school number of exam takers and the dropout rate do not change with conflict intensity. These results indicate that—among the enrolled students—those taking the exam at the end of grade 12 are not selected by the conflict.

In summary, we do not find evidence that higher conflict intensity induces lowerability students to continue to high school, to stay in school or to take the final exam, compared to other students. Taken together, these findings suggest that there is no evidence of a conflict-induced sample selection invalidating our analysis.

A second threat to our identification strategy is an omitted variable bias. This would occur if changes in some time-variant characteristics at the locality level were correlated with both conflict intensity and academic achievement. For instance, our identification strategy would be invalid if the geographical and temporal variations in conflict intensity were related to the economic conditions in the locality of the school. Available empirical evidence suggests that this is not the case. To begin with, our measure of conflict intensity, for example, the locality-level number of Palestinians fatalities, is not predicted by or correlated to local economic conditions. To test this formally, we estimate the following model:

Number of fatalities_{lt} =
$$a + \mathbf{Z}_{d,t-1} \times b + v_l + \partial_t + \mu_{dt} + u_{lt}$$
, (2)

where Number of fatalities_{lt} is the number of Palestinian fatalities in locality *l* in academic year *t*; $Z_{d,t-1}$ is a vector of previous year district-level economic conditions¹⁹ (average wage, low-skilled wage,²⁰ and unemployment rate); v_l and ∂_t are localityand academic-year fixed effects, respectively; μ_{dt} is a vector of district-specific linear time trends; and u_{dt} is the error term. In addition, we regress the number of fatalities on the same year district-level economic conditions. Results shown in Table A.6 of Online Appendix A indicate that the number of Palestinian fatalities at the locality level are not predicted by local economic conditions. Nonetheless, we cannot exclude that other time-varying unobservables may be driving our results. However, because the inclusion of the economic controls has little effect on the coefficient of *Pass exam* (compare columns (1) and (3) in Table 2), it is unlikely that results are driven by other unobserved economic variables.

A third possible threat to our identification strategy is that of reverse causality, that is, student's academic achievement determines conflict intensity at the locality level, with localities with more dropouts or low-performing students having a higher number of fatalities.²¹ In this case, the estimated coefficient for the number of fatalities would

^{19.} Economic variables are computed at the district level because the PLFS does not report the locality of the individual.

^{20.} We interpret this variable as a proxy for students' opportunity costs of being in school.

^{21.} As suggested by Rodriguez and Sanchez (2012), being a dropout or failing the final exam may increase rebellion in adolescents in a context of violent conflict. In turn, this may contribute to increase the number of demonstrations, the level of violence and, possibly, the number of the consequent conflict-related fatalities.

be upward biased to the extent that the error term is capturing the fact that students not passing the final exam would be those fighting against the IDF, thus creating the conditions for more fatalities. Although we cannot completely rule out this possibility, we argue that the effect is unlikely to be large. As discussed previously, our data show no correlation between conflict intensity and dropout rate at the locality or school levels. Moreover, our measure of conflict intensity is the total number of all fatalities occurring during the academic year up to the date of the exam, thus before the exam actually takes place.

A final threat to our identification strategy would result from students responding to an increase in conflict intensity by changing residential location. If migrating students were the higher-ability ones, the negative effect of the conflict on exam results would just be due to the decision of these students to move away from schools in high-conflict localities. The available empirical evidence suggests that this is not the case. In fact, internal and external mobility was severely restricted in the West Bank during the Second Intifada (World Bank 2007a; CIA 2008).²² Although we cannot directly test the determinants of student migration decisions, there is indirect empirical evidence showing that Palestinian mobility was extremely low during the period under consideration. PCBS (2009) documents that internal migration was "very negligible" (PCBS 2009, p. 93) after the outbreak of the Second Intifada. Between 1997 and 2007, the percentage of Palestinians who migrated internally decreased by 16%. Only 7.1% of the total population in the West Bank moved within the same district in this ten-year period. Looking at individuals aged 19 or younger, there has been a reduction by 8% in the number of movers during this period. Interestingly, the main motivations for internal migration during the period under analysis is marriage. Based on these results, we argue that, in the context of the Second Intifada, internal and external migration of students should not be a threat to our identification strategy.

A related concern is that students may decide to move to a different school in the same locality to avoid higher conflict intensity. Although this may be a common behavior in other conflict settings, there are two elements suggesting that this is less likely in the case of the West Bank during the Second Intifada. First, as we have seen, there is no evidence that schools more exposed to conflict suffer a larger reduction in the number of exam takers (see Table A.6 in Online Appendix A). Second, this strategy is unlikely to reduce the risk of being exposed to violence. In the case of the West Bank during the Second Intifada, this risk increases with traveling time (see Save the Children 2003; World Bank 2007a). Since moving to a different school is likely to significantly increase the latter,²³ it is doubtful that this would decrease the risk of being victim of a violent event (Akesson 2014).

^{22.} World Bank (2007a, p. 1) reports that "administrative restrictions, rooted in military orders associated with the occupation of the West Bank [...] are used to restrict Palestinian access to large segments of the territories [...]. Permit policies limit the freedom of Palestinians to [...] move about outside of their municipal jurisdiction".

^{23.} The possibility of school choice *within* localities is limited. MOEHE data indicate than 76% of localities have less than four high schools with only very large localities having a larger number of schools.

5.1.2. Robustness Checks. Our results are robust to several checks. To begin, we include as an additional control in regression (1) the number of fatalities during the previous academic year. Results shown in Table A.7 of Online Appendix A (column (1)) indicate that the lagged conflict-intensity variable is not significant whereas the magnitude and significance of *Number of fatalities* is unchanged with respect to our main specification.²⁴ This suggests that the number of fatalities occurring during the academic year of the exam is the relevant variable to include.

Next, we conduct a placebo test on the timing of the conflict events. We estimate regression (1) including the locality-level number of Palestinian fatalities in the next academic year. Results shown in Table A.7 of Online Appendix A (column (2)) indicate that the effect of the number of fatalities in the year after the one in which the student takes the exam is not significant whereas the coefficient for *Number of fatalities* is only slightly reduced.

Third, we check that our results are robust to possible spillover effects. To this end, we include as an additional control in regression (1) the total number of Palestinian fatalities in the district the school belongs to minus the number of fatalities in the locality of the school (to avoid double counting). This variable allows us to capture in a simple way the effects of conflict events taking place in localities that are near the school from a geographical and an administrative perspective. Results shown in Table A.7 of Online Appendix A (column (3)) indicate that including this additional control slightly increases the magnitude of the effect of the number of Palestinian fatalities at the locality level and its effect remains highly significant. The coefficient of the proxy for the spillover effect has the expected negative sign. This suggests that higher conflict intensity in neighboring localities has a potential negative impact on students' academic performance. Yet, the effect is not significant at conventional levels. We interpret this result as suggesting that the locality level is the relevant dimension to look at to capture the effects of conflict intensity on academic outcomes.

Finally, we check that our results are robust to the inclusion of controls for the intensity of the internal mobility restrictions imposed by the IDF.²⁵ To this end, we use the index presented in Cali and Miaari (2018). The index is a time-varying, locality-level variable that for each locality and each year counts the number of all mobility obstacles (i.e., physical barriers and checkpoints) within thirty minutes of travel time (by existing roads) from the locality's centroid, weighted by the inverse of their travel

^{24.} Since we use school-fixed effects and are clustering errors at the locality level, the fact that including previous-year conflict intensity does not affect the magnitude and significance of *Number of fatalities* suggests that our main result is robust to the threat of serial correlation.

^{25.} One of the distinctive features of the Israeli–Palestinian conflict is the presence of restrictions imposed by the IDF on the mobility of goods and people within the OPT as well as across the border with Israel. As for internal mobility restrictions, these include checkpoints, roadblocks, and physical barriers between villages within the OPT. Cali and Miaari (2018) present suggestive evidence that the placement of internal mobility restrictions is exogenous to the evolution of the Palestinian labour market and motivated by military-security consideration by the IDF.

time.²⁶ Results shown in Table A.8 of Online Appendix A, column (2) indicate that the effect of conflict intensity on the probability of passing the exam does not change with respect to our main specification (see Table A.8 in Online Appendix A, column (1)), whereas the effect of the mobility restrictions (as measured by the coefficient *Mobility Restriction Index*) is not significant at conventional levels. Next, to account for possible differences in the effect of mobility restrictions depending on the locality-level population size, we compute the *Congestion Mobility Restriction Index* as the product between the *Mobility Restriction Index* and the locality-level population.²⁷ Results shown in Table A.8 of Online Appendix A, column (3), confirm that our main result is unchanged.

5.2. Additional Results

5.2.2. Alternative Measures of Academic Achievement. To test for the effect of conflict on the total test score as an alternative measure of academic achievement, we estimate again regression (1) using for $Result_{islt}$ the individual *Total test score* at the exam.²⁸ The effect of conflict intensity is significant and negative (Table 3, column (1)). The size of the estimated coefficient indicates that a one standard deviation increase in the number of fatalities decreases *Total test score* by 4 percentage points $[(-218.4/1000) \times 17.2]$, which, given that the mean value of the dependent variable is 640, is equivalent to a 1% reduction. This translates into a 2 points reduction in the *Total test score* for a student enrolled in a school in a locality characterized by the average level of conflict intensity $[(-218.4/1000) \times 7.7]$ relative to a student enrolled in a school in a locality that experienced no conflict. The reduction can be as large as 22 points for students in schools situated in the most conflict-exposed localities.

In columns (2) and (3), we report the results for the two most important subjects that are present in the exam for both streams, namely mathematics and Arabic. The results indicate that the effect of conflict intensity is larger for *Math test score* than for *Arabic test score*, with a one standard deviation increase in the number of fatalities decreasing the former by 0.03 standard deviations and the latter by 0.02. The larger effect for *Math test score* is in line with previous studies showing that math test scores are more sensitive to conditions in the learning environment (e.g., Hanushek and Rivkin 2010; Monteiro and Rocha 2017; Gershenson and Tekin 2018).²⁹

^{26.} The weight captures the idea that the more distant a mobility obstacle is (taking into account the road system) the less it affects mobility to and from the locality. For additional detail on the index, see Cali and Miaari (2018).

^{27.} The basic idea here is that for a given number of barriers, the larger the population of a locality the more time-consuming it would be to pass the barriers due to the larger number of people to be checked by the IDF.

^{28.} See Online Appendix B for details on how the *Total test score* is computed for the Adabi and the Elmi streams, respectively.

^{29.} These results are of comparable magnitude to those from previous studies looking at the impact of conflict and violence on academic achievement, even if referred to samples and school grades different

	Total test score (1)	Math test score (2)	Arabic test score (3)	Access University (4)	Access University Abroad (5)
Number of fatalities	-218.43**	-50.44***	-22.99***	-0.53***	-0.12
	(93.24)	(15.10)	(7.43)	(0.18)	(0.18)
Individual characteristics	Yes	Yes	Yes	Yes	Yes
District-level economic controls	Yes	Yes	Yes	Yes	Yes
Birth-cohort fixed effects	Yes	Yes	Yes	Yes	Yes
School-fixed effects	Yes	Yes	Yes	Yes	Yes
Academic-year fixed effects	Yes	Yes	Yes	Yes	Yes
Number of observations	138,135	138,060	138,135	138,135	138,135
Mean of the dependent variable	640.5	64.3	66.8	0.53	0.17

Notes: OLS regression results for equation (1). *Total test score* is the standardized total test score at the exam (see Online Appendix B for details). *Math test score* is the standardized test score for the math module of the exam. *Arabic test score* is the standardized test score for the Arabic module of the exam. For details on how *Access University* is a dummy variable that takes value 1 is the student total score at the exam is higher than the minimum test score necessary for admission to university (650) and 0 otherwise. *Access University Abroad* is a dummy variable that takes value 1 if the student total score at the exam is higher than the minimum test score necessary for admission to university (650) and 0 otherwise. *Access University Abroad* is a dummy variable that takes value 1 if the student total score at the exam is higher than the minimum test score necessary for admission to university (650) and 0 otherwise. *Mumber of fatalities* is the locality-level number of conflict-related Palestinian fatalities caused by the IDF during the academic year divided by 1000. *Individual characteristics* include: *Male* (a dummy that takes value 1 if the student is male and 0 otherwise); *Christian* (a dummy that takes value 1 if the student is errolled in the Adabi stream and 0 if enrolled in the Elmi stream). *District-level economic controls* are the same as in Table 2. In each regression, a constant is included but not shown. Robust standard errors in parenthesis are clustered at the locality level. **Significant at 5%; ***significant at 1%.

Finally, we consider two other measures of academic achievement. We estimate model (1) using for *Result_{islt}*, respectively: *Access University*, a dummy variable that takes value 1 if the student obtains at least the minimum test score for admission to university in the OPT (i.e., 650 over 1000) and 0 otherwise; and *Access University Abroad*, a dummy variable that takes value 1 if the student gets at least the minimum test score required for admission to university in other Middle Eastern countries and the engineering major in Palestinian universities (i.e., 850 over 1000) and 0 otherwise. Results reported in column (4) indicate that the conflict reduces students' probability of achieving the minimum test score required for admission to university in the conflict does not reduce the probability.

from ours. Monteiro and Rocha (2017) look at elementary schools in Brazil and report a 0.010 standard deviation reduction in math test scores in schools exposed to drug-related violence. Instead, the effect of violence on reading test scores is never significant. Kibris (2015) looks at the university admission test in Turkey in 2005 and finds that each casualty is associated with a 0.014 and 0.013 points decrease in the math and Turkish test scores, respectively. In our case, each fatality reduces the math test and the Arabic test score by 0.05 and 0.02 points, respectively. Finally, Gershenson and Tekin (2018) show that primary schools within a five miles radius of a sniper attack have a reduction between 2% and 4% in the *school-level* pass rate in math. Computing a similar measure for our analysis, we find that a one standard deviation increase in the number of fatalities leads to a 2% reduction in the probability of passing the math part of the exam.

of achieving the minimum test score necessary for admission to universities abroad. We interpret these results as suggesting the possibility that the effect of conflict is heterogeneous across levels of academic ability, with high-ability students possibly being less affected by the conflict. We explore this issue in more detail in what follows.

5.2.3. Heterogeneity. Performance at the Exam: Conflict intensity reduces the Total test score at the exam. Yet, it is possible that the magnitude of the effect is heterogeneous along the Total test score distribution. To test for this, we employ a quantile regression model (Koenker and Hallock 2001). Results are shown in Table 4, columns (2)–(6). Column (1) reports for comparison the OLS estimate of the effect of conflict intensity on Total test score (the same as in Table 2, column (1)). The quantile regression results indicate some interesting differences in the effect of conflict across the various parts of the conditional distribution of the Total test score. The conflict lowers all the quantiles of the distribution but the 95th, for which the effect of conflict on Total test score is instead very small and not significant.³⁰ Under the assumption that conflict does not affect the rank of students in the test score distribution, these results indicate that higher conflict intensity, whereas it reduces both the probability of passing the exam (see Table 2) and of being admitted to university (see Table 3), has no effect on students in the upper tail of the distribution.

Gender of the Student Previous research has documented that exposure to conflict can result in significant gender differentials in individual educational outcomes (Chamarbagwala and Morán 2010; Shemyakina 2011; Buvinić et al. 2014; Valente 2013). To check for this possibility in the context of the Second Intifada, we include the interaction between the number of fatalities and the dummy for the gender of the student (*Male*). Results reported in Table A.9 of Online Appendix A indicate that there is no evidence of a statistically significant differential effect of conflict intensity between boys and girls on the probability of passing the exam or on the total test score. This suggests that the mechanisms behind the impact of conflict on academic achievement are gender neutral.

Stream of Study, Age Cohort, and Private versus Government Schools In unreported regressions (available upon request), we also look at the possible differential effects of conflict by stream of study, by age cohorts, and by school supervising authority (government/private).³¹ These additional checks do not alter our conclusions in any meaningful way.

^{30.} In fact, quantile coefficients are not significantly different at conventional levels among them but for the 95th quantile.

^{31.} Conflict intensity does not influence the stream (Adabi vs. Elmi) or the type of school (government vs. private) the student enrols in. There is no evidence that the conflict has a differential effect on the probability of passing the exam in either of the two streams, whereas its effect is larger and it is significant only for students in government schools.

lts.
resu
egression
l quantile 1
anc
OLS
score:
test
total
on
intensity
conflict
t of
Effect
TABLE 4.

				101al 16	est score			
	OLS				Quantiles			
	(1)	5th (2)	10th (3)	25th (4)	50th (5)	75th (6)	90th (7)	95th (8)
Number of fatalities	-218.43^{***} (93.24)	-200.60^{***} (83.77)	-239.98** (90.47)	-204.93^{**} (65.42)	-188.93^{**} (58.15)	-240.57^{***} (55.47)	-178.43^{**} (56.59)	-46.68 (49.35)
All controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Birth-cohort fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
School-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Academic-year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. observations	138,135	138,135	138,135	138,135	138,135	138,135	138,135	138,135
Mean of dep. variable	640	220	330	522	670	807	890	927
Notes: Column (1) reports OLS regr fatalities is the number of all conflic	ression results. Colu	umns (2)–(6) repoi n fatalities caused	rt quantile regres by IDF at the loc	sion results. The cality level durin	dependent variab g the academic ye	le for all regression ar divided by 100	ons is <i>Total test se</i> 00. All controls inc	ore. Number of slude Individual
<i>characteristics</i> and <i>District-level ecc</i> are clustered at the locality level. For	onomic controls as 1 r columns (2)–(6), b	n Table 3. In each ootstran standard	regression, a cor errors in narentle	stant is included	l but not shown. F	or column (1), rot ficant at 1%.	oust standard erro	s in parenthesis

6. Possible Transmission Mechanisms

6.1. Changes in the Quality of the School-Level Learning Conditions

The first mechanism we consider is that of the conflict-induced change in the quality of the learning environment at school. To show that this may contribute to explain the negative effect of conflict on academic achievement, we first provide evidence of the effect of conflict on a set of school characteristics that determine the quality of the learning environment. Next, we show how those characteristics are correlated with exam performance.

We proxy school characteristics using a large set of variables provided by the MOEHE. These are: *Student/Teacher Ratio, Average Size of the Classroom, Average Density in the Classroom,* and *Teachers' Average (years of) Experience.* The *Student/Teacher Ratio* is the school-level ratio between the number of students and the number of teachers. The *Average Size of the Classroom* is the school-level average size of the classroom measured in square meters, that is, it is a measure of the *physical size* of the classroom. *Teachers' Average Experience* is the average number of teaching years of the teachers in the school. The *Average Density in the Classroom* is the average number of students per square meter in the classroom. To these variables commonly used in the literature to proxy for the quality of school supply, we add the *Percentage of Christians in the School,* to proxy for religious and cultural heterogeneity at the school level.

To estimate the effect of conflict intensity on the quality of the school-learning environment, we regress each school characteristic on the locality-level number of Palestinian fatalities caused by the IDF during the academic year, controlling for school- and academic-year fixed effects.

Results are reported in Table 5. Conflict intensity increases *Average Density in the Classroom* (column (1)). There are two main reasons for this effect. First, military attacks and school occupations by IDF damaged and—in some cases—destroyed premises and properties of Palestinian schools (Palestinian Ministry of Education and Higher Education (MOEHE) 2005). Several classrooms became unavailable due to the destruction of walls, doors, windows and furniture (Giacaman et al. 2002). Military operations during the Second Intifada indeed caused a chronic shortage of classrooms (UN-OCHA-OPT Humanitarian Monitoring, various issues).³² Second, the conflict made it more difficult for the Palestinian Authority to increase the supply of schools in the OPT to meet the growing number of school-aged Palestinians (World Bank 2002).³³ The need to rebuild infrastructure damaged by the conflict reduced

^{32.} Additional regressions using the MOEHE data set (available upon request) show that the conflict reduces the *Number of classrooms* and the *Size of the school flat*, even if both effects are not significant at conventional levels.

^{33.} The number of students in the OPT has been constantly increasing since the creation of the national curriculum with the 1994 Oslo Agreements. In fact, overcrowding was also a problem before the Second Intifada due to school enrolment growing at the rate of 2% annually and serious difficulties in obtaining

	Density in the classroom (1)	Students/ teacher ratio (2)	Teachers' average experience (3)	Average size of the classroom (4)	Percentage of Christians (5)
Number of fatalities	2.66*** (0.50)	-25.88*** (6.77)	12.14*** (3.77)	10.64 (8.15)	0.13 (0.13)
School-fixed effects Academic-year fixed effects	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes
Number of observations	2651	2643	2649	2651	2651
Mean value of the dependent	1.5	18.1	9.6	39.5	0.1

TABLE 5. Effect of conflict intensity on the quality of the school learning conditions.

Notes: OLS regression results. The dependent variable for each regression is reported in the first row of the table. *Number of fatalities* is the locality-level number of conflict-related Palestinian fatalities caused by the IDF during the academic year divided by 1000. *Average Size of the Classroom* is the average number of students per square meters in the classroom. *Students/Teacher Ratio is the school-level ratio between the number of students and the number of teachers. Teachers' Average Experience is the average number of teaching years of the teachers in the school. Average Density in the Classroom is the school-level average size of the classroom measured in square meters. Percentage of Christians is the school-level percentage of Christian students. In each regression, a constant is included but not shown. Robust standard errors in parenthesis are clustered at the locality level. ***Significant at 1%.*

the resources available to expand existing schools or to construct new ones. In fact, during the Second Intifada, the development or upgrading of Palestinian schools was very difficult, even when financial resources were available. Since permits for public infrastructure projects in the West Bank were managed by the Israeli Government, investment projects were delayed or even refused during the conflict period (World Bank 2008). All these factors contributed to overcrowded classrooms, creating a serious challenge to the Palestinian education system and to students' learning (World Bank 2002; UNESCO 2011b).

Results shown in Table 5 also indicate that conflict intensity reduces *Student/Teacher Ratio* (column (2)) and increases *Teachers' Average Experience* (column (3)). These results suggest that not all aspects of the school learning conditions worsen with an increase in conflict intensity. In fact, a lower student/teacher ratio is usually believed to improve academic achievement (Alderman et al. 2001). The same positive impact is expected from longer teacher experience, although typically studies have found that there is a positive effect for the first three years of experience and any additional year has little effect (Rivkin et al. 2005). Although there may be different explanations for these results, one possibility is that MOEHE compensated schools hit

construction permits in the OPT from Israel (Sultana 2003). The conflict, by damaging existing schools and reducing the number of permits to expand them, has probably exacerbated an already existing problem.

by violence by sending more teachers³⁴ and teachers with more experience.³⁵ Finally, conflict intensity does not affect the *Average Size of the Classroom*³⁶ (column (4)) nor the composition of the student pool (at least in terms of religion), as measured by the *Percentage of Christians in the School* (column (5)).

Next, we test whether these school characteristics are correlated with student performance at the final exam. To this end, we estimate the following model:

$$Pass \ exam_{islt} = a + S'_{slt} \times \vartheta + \mu_c + \gamma_s + \partial_t + u_{islt}, \tag{3}$$

where *Pass exam_{islt}*, is a dummy variable that takes value 1 if the student passes the final exam and 0 otherwise and S'_{slt} includes the following school characteristics: *Student/Teacher Ratio, Average Size of the Classroom, Average Density in the Classroom, Teachers' Average Experience* and *Percentage of Christians in the School*. As in model (1), we also include a full set of birth-cohort (μ_c), school- (γ_s), and academic-year (∂_t) fixed effects.

Results shown in Table A.10 of Online Appendix A indicate that the probability of passing the exam is negatively correlated with *Average Density in the Classroom* (our proxy for difficulties in teaching and learning conditions in the classroom),³⁷ whereas it is positively correlated with the *Average Size of the Classroom* (our proxy for quality of school infrastructures).³⁸ Interestingly, the other two school characteristics that are affected by conflict (see Table 5), namely *Student/Teacher Ratio* and *Teachers' Average Experience*, are not correlated with the probability of passing the exam. Finally, results show that the proxy for religious heterogeneity, that is, *Percentage of Christians in the*

^{34.} The conflict-induced decrease in the *Student/Teacher Ratio* is indeed the result of the combination of the (nonsignificant) effect of conflict on the number of high school students (negative, see Section 4) and on the number of teachers (positive, results available upon request).

^{35.} An alternative interpretation for the increase in the *Teachers' Average (years of) Experience* are the difficulties faced by the MOEHE in recruiting fully qualified staff during the Second Intifada (Save the Children 2003). In this case, a higher *Teachers' Average (years of) Experience* would not proxy for higher teaching staff capabilities to prepare the students for the exams, but it would more likely capture an increasingly older teaching staff.

^{36.} Note that the result that conflict intensity does not affect the Average Size of the Classroom does not contradict our explanation of the effect of conflict on Average Density in the Classroom. The fact that the conflict destroys some classrooms (or makes them unusable due to damage) does not imply that the average size of the existing classrooms needs to change as well. Moreover, the two measures capture different aspects of the quality of the school: whereas Average Size of the Classroom can be interpreted as a proxy for the quality of school-level infrastructures, Average Density in the Classroom is instead a proxy for the quality of the teaching and learning conditions in the classroom.

^{37.} Although the literature on the relationship between overcrowding and student learning is quite limited (McMullen and Rouseb 2012), there is some evidence that overcrowding can have an adverse impact on learning. Rivera-Batiz and Marti (1995) show that in New York public schools between 2% and 9% fewer students pass reading and mathematics proficiency tests in overcrowded schools compared to students in nonovercrowded schools.

^{38.} Even though there is no consensus on which specific characteristics matter most, several studies have documented the importance of the quality of school infrastructures and of classroom design for the learning process (Higgins et al. 2005; Cleveland and Fisher 2014).

school, is not relevant for explaining academic achievement for students in the West Bank.

Reading these results together with those from Table 5 provides two interesting insights. First, the effect of conflict on exam results is attenuated when school learning conditions are controlled for. Indeed, when we include the full set of school characteristics in regression model (1), the coefficient for *Number of fatalities* becomes 0.45, which is 14% smaller than in the main specification (results available on request). Second, they indicate that, among all school characteristics, *Average Density in the Classroom* is the only one affected by conflict that is also correlated with the probability of passing the exam. We interpret this as suggestive evidence that the conflict-induced increase in classroom density is a possible transmission mechanism explaining the negative impact of the conflict on students' results at the final exam. Our computations indicate the deterioration in school learning conditions can indeed account for 23% of the total effect of conflict intensity on the probability of passing the exam. Interestingly, this result is in line with the findings of reports and case studies showing that one of the major barriers to the provision of quality education in the OPT is the low quality of school infrastructures and classrooms (UNDP 2015).

This conclusion is reinforced by the results emerging from the analysis of the effect of school characteristics along the test score distribution. Quantile regression analysis (results available on request) reveals that the effects of *Average Density in the Classroom* and *Average Size of the Classroom* are significant only for students in the lower tail of the test score distribution. This suggests that those are the students for whom the quality of the learning environment at school is likely to be more important.

6.2. Changes in Students' Psychological Well-Being

There is abundant evidence suggesting that the student's psychological well-being is an important determinant of academic achievement (Roeser et al. 1998). In particular, previous research has documented the negative relationship between cognitive performance and exposure to violence (Wilson and Rosenthal 2003; Sharkey 2010). Youth directly or indirectly exposed to different forms of violence show reduced sleep, increased anxiety and difficulties concentrating, possibly leading to reduced academic performance (Pynoos et al. 1987; McGill et al. 2014; Gershenson and Tekin 2018). In fact, the psychological impact of the security conditions is an important element to be considered to understand the determinants of Palestinian students' academic performance (World Bank 2006).

Although our data set does not allow for a direct test of these effects,³⁹ in the following we provide indirect evidence for the possible role of the psychological mechanism in explaining our main result. As a first piece of evidence, we estimate model (1) using as a measure of conflict intensity the *per-capita* number of Palestinian

^{39.} To the best of our knowledge, the only two studies that empirically test the psychological effects of conflict are Do and Iyer (2012) and Swee (2015), both looking at the relationship between exposure to conflict and depression.

		Pass exam	
	(1)	(2)	(3)
Per-capita number of fatalities	-12.44^{***} (4.51)		
Per-capita number of young fatalities		-88.22*** (25.56)	
Per-capita number of non-young fatalities		-1.73 (5.94)	
Per-capita number of Palestinians killed while not fighting			-16.37^{***} (6.19)
Per-capita number of Palestinians killed (other)			-1.42 (11.69)
Individual-level characteristics	Yes	Yes	Yes
District-level economic conditions	Yes	Yes	Yes
Birth-cohort fixed effects	Yes	Yes	Yes
School-fixed effects	Yes	Yes	Yes
Academic-year fixed effects	Yes	Yes	Yes
Number of observations	135,090	135,090	135,090
Mean of the dependent variable	0.74	0.74	0.74

TABLE 6.	Effect of direct	exposure to conflic	t on the	probability	of	passing	the fina	l exam

Notes: OLS regression results for equation (1). The dependent variable is *Pass exam*, a dummy that takes value 1 if the student has passed the final exam and 0 otherwise. *Per-capita number of fatalities* is the per-capita number of Palestinian fatalities caused by IDF at the locality level computed by dividing the number of fatalities at the locality level occurred during the academic year by the locality-level population in 1997. *Per-capita number of Palestinians killed (other)* include all those killed excluding those that were not fighting, that is, those fighting and those for which the status is unknown. See the text for additional details on the variables construction. *Individual characteristics* and *District-level economic controls* are the same as in Table 3. In each regression, a constant is included but not shown. Sample is smaller than in Table 2 because of missing values for the variable *Population at the locality level in 1997*. Robust standard errors in parenthesis are clustered at the locality level. ***Significant at 1%.

fatalities at the locality level during the academic year.⁴⁰ We use this measure to proxy for the intensity of the *direct exposure* to violence experienced by the student: the higher the value of the *per-capita* number of fatalities, the higher the probability that the student has been (physically) close to a violent event.

Results are shown in Table 6, column (1). Direct exposure to fatalities has a negative and highly significant effect on academic achievement. This can be interpreted as indicating that the more student is (or believes she is) likely to be the victim of a conflict event, the lower her academic performance is. Next, we consider the possibility that a process of self-identification by the student with young fatalities contributes to the effect of conflict intensity on academic performance. In column (2), we use as a

^{40.} The *per-capita number of fatalities* at the locality level is computed dividing the number of fatalities at the locality level occurred during the academic year divided by the locality-level population in 1997, which is the only pre-Second Intifada year for which population data at the locality level are available.

measure of conflict intensity the *per-capita number of young* (less than 18-year-old) *Palestinian fatalities* in the locality during the academic year. Results show that the effect is highly significant and larger in magnitude than the effect of *per-capita number of non-young Palestinians fatalities*, with the difference between the two coefficients being statistically significant at 1%. Interestingly, these findings are in line with the psychological literature showing that the type of conflict-related violence to which the individual is exposed to is a key factor driving the incidence of both mental health disorders and violence-related trauma symptoms (RAND 2015).

Finally, we construct a measure of conflict intensity looking at the characteristics of the fighting status of the killed. In column (3), we proxy conflict intensity using the *per-capita number of Palestinians killed by the IDF while not taking part in the hostilities*.⁴¹ The negative impact of conflict intensity as measured by those killed while not fighting is negative, highly significant and larger than the effect of other fatalities (those killed while fighting or unknown status). Yet, we cannot reject the null hypothesis that the two coefficients are equal.

To provide additional evidence supporting the psychological mechanism, we look at the impact of the direct exposure to conflict at various points in time during the academic year. In particular, we are interested in the effect of conflict events occurring very close in time to the date of the exam. The basic idea is that—if significant— the negative effect of these events on student's exam result would be difficult to explain by a worsening in the quality of the school learning environment.⁴² Instead, the effect would be consistent with the direct exposure to conflict negatively affecting the student's psychological well-being and—in turn—the performance at the final exam.

To this end, we take advantage of the information on the exact date of each fatality included in the B'Tselem data set and compute the per-capita number of fatalities occurred in each locality 1 month before the exam, 2 months before the exam and so on, up to 9 months before the exam. Then, we re-estimate our main regression model (1) including all these measures of direct exposure to conflict at the same time. This allows us to provide a detailed characterization of how the effect of direct exposure to conflict varies depending on the specific moment at which it occurs with respect to the date of the exam.

^{41.} B'Tselem defines *Palestinians killed by the IDF who was not taking part in the hostilities* as people not participating directly in hostilities at the time they were killed and were not fulfilling a continuous combat function. Examples taken from the B'Tselem data set are: [name omitted] 32 years old, killed near Ramallah by gunfire when hiking with relatives on hills. Without warning, soldiers opened fire at them from about 500 m away. [name omitted] 15 years old, killed near Kfar Dan by gunfire. Shot in the back by Border Police who passed by in a jeep. Testimony given to B'Tselem indicates the street was quiet and the officers were not in danger. [name omitted] 25 years old, killed near Nablus by gunfire while serving customers in his butchery.

^{42.} Conflict-induced changes in the quality of the school learning environment take time to materialize. Moreover, it is unlikely that those school-level changes would immediately affect students' academic performance.



FIGURE 1. Effect of direct exposure to conflict on the probability of passing the final exam by months before the exam. Dependent variable is the probability of passing the final exam (*Pass exam*). The Figure plots the estimated coefficients for direct exposure to conflict in the *n*th month before the exam, as measured by the per-capita number of fatalities caused by IDF *n* (with n = 1, ..., 9) months before the exam at the locality level. The solid vertical lines show the 90% confidence interval of each estimate, whereas the dashed horizontal line indicates zero.

Figure 1 depicts the estimated coefficients and the corresponding confidence intervals for the per-capita number of fatalities at the locality level in each month. Results indicate that direct exposure to conflict events in any of the months of the academic year has a negative effect on the probability of passing the final exam. Yet, the magnitude of the effect is small and not precisely estimated in most cases. Interestingly, the effect is significant and the largest for the conflict events occurring 1 month before the exam. Although events occurring earlier also seem to matter (see for instance the coefficient of the lag at 6 months), we interpret the large negative effect associated with being exposed to conflict-related violence shortly before the exam as providing support for the psychological mechanism.

These results are in line with psychological research that has documented conflict exposure as the origin of high levels of emotional and behavioral problems and of psychological disorders in adolescents in the OPT (Dubow et al. 2009; Mataria et al. 2009; Thabet et al. 2002). Palestinian adolescents exposed to conflict-related violence report higher levels of depression and mental disorders, Post-Traumatic Stress Disorders (PTSD) symptoms (Al-Krenawi et al. 2007; Abdeen et al. 2008; RAND 2015) and an increase in symptoms like restlessness, fear, anxiety, communication difficulties and a reduced ability to concentrate on their studies (UNRWA 2003; Arafat

and Boothby 2003; PCBS 2004; Save the Children 2004). Pedersen et al. (2001) report that heavy shelling and attacks directly affect students' concentration on schoolwork and makes final exam preparation more difficult. Giacaman et al. (2007) find that the experience of conflict-related violence is a strong predictor of depression-like states that in turn may affect academic performance.⁴³ Taken together, this evidence supports the view that the psychological channel plays a role in explaining the negative effect of the conflict on the probability of passing the final exam.

One difference between the psychological mechanism and the school-quality mechanism is that, although the latter is geographically localized, the former is much less so. Young Palestinians are exposed to conflict-related violence not only through direct or indirect experience but also through intense media coverage. As reported by Save the Children (2003), children often talk about conflict-related incidents seen on television. In addition, it is common for images of Palestinians killed to be displayed in public places, including school walls (Sultana 2006). Since the West Bank is a small geographical area and there are more than 48 private radio and television stations, all the information related to the conflict and the political situation rapidly and fully spreads among the population (European Union 2006). The intense media coverage of the conflict is likely to impact on students' psychological well-being. Gvirsman et al. (2014) show that higher levels of exposure to media covering violent events related to the conflict are associated with higher levels of PTSD symptoms in young Palestinians. The larger this media effect is, the more likely our results are to represent a lower bound of the effect of conflict on academic achievement.

6.3. Other Possible Mechanisms

6.3.1. Days of School Closure. Another possible mechanism explaining our main result could be the negative effect of conflict intensity on the quantity of learning time at school, the latter being an important determinant of students' academic performance. Anecdotal evidence indicates that during the Second Intifada the Palestinian Authority closed schools and interrupted the normal schooling activities for security reasons.⁴⁴ Often, these decisions were the response to episodes of occupation of schools by the IDF or to situations of conflict-related violence around the schools (MOEHE 2005; UN-OCHA-OPT various issues).

To test the effect of the reduction in learning days due to conflict on students' performance at the final exam, we use confidential data provided by the MOEHE on the percentage of school days lost due to school closures for security or military reasons.

^{43.} Pedersen et al. (2001) report the interview of a school principal saying: "When the students come to school they are very often afraid, sometimes they have not slept because there has been shooting or bombing during the night [...]. Sometimes the Israeli soldiers shoot after the students when they go to school [...]."

^{44.} Save the Children (2003) reports that 580 schools closed due to Israeli military imposition in 2002. MOEHE (2005) reports that 1,135 school days were lost and 850 schools faced temporary closure during the Second Intifada.

Unfortunately, these data suffer from two limitations. First, data are available only at the district level. Second, data are not available for the whole period under analysis. In particular, there are no data for the 2000–2001 and the 2001–2002 academic years. We estimate the impact of conflict intensity on students' learning time by regressing *Percentage of days lost due to closure of school* (i.e., the district-level percentage of days lost due to closure of school (i.e., the district-level percentage of days lost due to closure of school sfor security/military measure) on district-level conflict intensity, controlling for district-level economic conditions and district- and year-fixed effects. Table A.11 of Online Appendix A shows that conflict intensity increases the percentage of school closure days, that is, it reduces the number of learning days at school.

Next, we check the possible role of *Percentage of days lost due to closure of school* as a mechanism behind our main result. To this end, we include it in regression (3) together with the school characteristics discussed in Section 6.1. Results shown in Table A.12 of Online Appendix A report that its coefficient is negative, suggesting that a reduced number of days of school is associated with a lower probability of passing the exam. Yet, the effect is not significant at conventional levels.⁴⁵ Interestingly, the negative effect of *Average Density in the Classroom* on *Pass exam* remains highly significant and its magnitude does not change with respect to that reported in Table A.10 of Online Appendix A.

6.3.2. Student Absenteeism. Conflict may reduce academic achievement by increasing student absenteeism (World Bank 2006; Giacaman et al. 2007). Unfortunately, the possibility of testing this mechanism in our setting is limited by the absence of either individual-level or school-level information on student absenteeism. To gain some insights into the possible role of student absenteeism in explaining our main result, we use confidential data provided by the MOEHE on the districtlevel percentage of students who lost days of school due to the conflict. Yet, these data cover only two academic years (2000–2001 and 2001–2002). Results indicate a positive correlation (0.4, significant at 1%) between district-level conflict intensity and percentage of students who lose days of schools, that is, suffering a reduction in learning days. However, the number of observations is very small and one should thus be cautious in interpreting this evidence. Finally, we note that according to MOEHE high school regulations, students enrolled in high school who have been absent for more than 30 days during the academic year without justification are not allowed to take the final exam (Nicolai 2007). This implies that students with high absenteeism are not included in the MOEHE data set we use in our main analysis.

^{45.} When we control for *Percentage of days lost due to closure of school* in our main regression (1), the effect of the *Number of fatalities* is reduced by 4%. The coefficient for *Percentage of days lost due to closure of school* is negative but not significant (results available upon request). Note that when we control for *Percentage of days lost due to closure of school* we lose 31% of our observations because of missing values. In this restricted sample, the coefficient for the *Number of fatalities* is twice as large as in our baseline regression (-1.15, significant at 1%).

6.3.4. Quality of Teaching and Teacher Absenteeism. Another channel through which conflict may reduce students' academic performance is via its effect on teachers (Hanushek and Woessmann 2011, Chap. 2). There is abundant anecdotal evidence that the conflict affected teachers and teaching activities in the West Bank. Teachers had to cope with repeated rights violations and the negative psychological impact of pervasive violence in their lives (Nicolai 2007). As reported by a school principal in Pedersen et al. (2001, p. 18): "several of our teachers have martyrs and injured in their families". Giacaman et al. (2002) note that teachers face problems related to frequent class interruptions, cancellation of school days, increases in workload and management of traumatized children. Although there are few doubts that conflict affects daily teacher activities, Giacaman et al. (2002) argue that the effect of conflict on teaching quality is not obvious in the context of the Second Intifada. There is some evidence showing that the difficult situations experienced by the teachers had both positive and negative effects on the quality of their teaching. In fact, although some teachers reported being chronically exhausted with a decline in their performance, others experienced a sense of increased commitment, covering other delayed or absent teachers and providing special care to the students (Nicolai 2007).

Previous research has shown that teacher absenteeism—by reducing the quantity of learning time—is another possible mechanism behind the negative effect of conflict and violence on education (Justino, Marinella and Salardi 2014; Monteiro and Rocha 2017). Unfortunately, MOEHE does not collect individual or school-level data on teacher absenteeism. The only existing data are the district-level percentage of teachers losing at least one day of school for military reasons. These data are available only for two years (2000–2001 and 2001–2002). Results indicate a positive correlation (0.3, significant at 10%) between district-level conflict intensity and the percentage of teachers missing days of school. However, since we have observations only for two years, we cannot include this variable as an additional control in our regression analysis of the impact of school-level characteristics on academic achievement (equation (3)).

Although not conclusive, our evidence is line with the case studies reported in Save the Children (2003) showing that teachers faced severe difficulties in accessing schools during the Second Intifada. In fact, the type and range of problems teachers experience because of the conflict are not different from those reported by other Palestinian workers: delays in reaching the workplace because of checkpoints, risk of being exposed to shootings, and so forth. Yet, the education sector is reported to have encountered the fewest problems with absenteeism among all public sectors (World Bank 2007a).

Taken together, these results seem to indicate that there is some evidence that conflict intensity is associated with more school closure days and more student and teacher absenteeism. Yet, severe limitations on data availability do not to allow us to establish robust empirical relationships between these variables and student performance at the exam. Therefore, we cautiously interpret the overall evidence on these mechanisms as inconclusive. 6.3.5. Parental Distress. Finally, we consider as a potential mechanism the fact that conflict may negatively affect the well-being of parents, causing them—in turn—to have a negative influence on their children's academic performance. Several psychological studies suggest the possibility that adults experiencing conflict-induced stress modify their behavior toward their own children. Ayer et al. (2017) argue that nearly all psychological studies looking at the Israeli–Palestinian conflict find that individuals more exposed to violence are more likely to be unable to carry out normal functions related to personal relationships. Thabet et al. (2008) and Srour (2005) document that exposure to war trauma impacts on both parents' and children' mental health and that their emotional responses are inter-related.

Although these studies suggest the possible existence of a parental distress channel affecting student academic performance, we cannot test this hypothesis with our data. Yet, as long as parental exposure to conflict can be proxied by direct exposure to conflict experienced by the student, we argue that this mechanism is captured by the effect of the *per-capita number of fatalities* on the probability of passing the final exam, as discussed in Section 6.2.

7. Concluding Remarks

In this paper, we analyze the impact of violent conflict on the academic achievement of high school students. In particular, we study the effect of the Second Intifada on the individual results of the high school leaving exam (Tawjihi General Examination) for the whole population of Palestinian students enrolled in the Adabi (Literary) and Elmi (Scientific) streams in the West Bank during the period 2000–2006.

Our results show that conflict reduces the probability of passing the final exam, the total test score and the probability of being admitted to university. We also document that the magnitude of the effect of conflict varies along the test score distribution and with the timing and the type of violent events a student is exposed to.

Several transmission mechanisms could explain our findings. First, we show that the conflict negatively affects the quality of the school learning environment and that this in turn negatively correlates with results on the final exam. Second, we provide suggestive evidence consistent with a psychological mechanism contributing to the explanation of our main result. In particular, we show that direct exposure to conflict has a large negative impact on the probability of passing the exam and that the effect is the largest for conflict events occurring shortly before the exam date. Together, these results indicate that the conflict has a negative impact on exam results by affecting both the learning process and performance at the exam. The fact that the two mechanisms we identify have very distinct characteristics implies that effective education policies to mitigate the negative effects of conflict on academic achievement need to account for each of them.

Finally, we note that our findings indicate that the negative effect of conflict on academic achievement may also have long-lasting consequences. By reducing the probability of students successfully completing their high school studies and of being admitted to university, conflict may significantly hinder human capital accumulation and economic development. This is a potential long-term cost to be added to the several other already documented negative consequences of the Israeli–Palestinian conflict.

References

- Abdeen, Ziad, Radwan Qasrawi, Shibil Nabil, and Mohammad Shaheen (2008). "Psychological Reactions to Israeli Occupation: Findings from the National Study of School-Based Screening in Palestine." *International Journal of Behavioral Development*, 32, 290–297.
- Akbulut-Yuksel, Mevlude (2014). "Children of War: The Long-Run Effects of Large-Scale Physical Destruction and Warfare on Children." *Journal of Human Resources*, 49, 634–662.
- Akesson, Bree (2014). Contradictions in Place: Everyday Geographies of Palestinian Children and Families Living Under Occupation. McGill University, Montreal.
- Al Kafri, Saleh (2004). *Palestinian Labor Market Demand Changes for Educational Outcomes under the Israeli Occupation*. PCBS, Ramallah.
- Al-Krenawi, Alean, John R. Graham, and Mahmud A. Sehwail (2007). "Tomorrow's Players under Occupation: An Analysis of the Association of Political Violence with Psychological Functioning and Domestic Violence, among Palestinian Youth." *American Journal of Orthopsychiatry*, 77, 427–433.
- Alderman, Harold, Peter F. Orazem, and Elizabeth M. Paterno (2001). "School Quality, School Cost, and the Public/Private School Choices of Low-Income Households in Pakistan." *The Journal of Human Resources*, 36, 304–326.
- Amodio, Francesco and Michele Di Maio (2018). "Making Do with What You Have: Conflict, Input Misallocation, and Firm Performance." *Economic Journal*, 128, 2559–2612.
- Arafat, Cairo and Neil Boothby (2003). A Psychological Assessment of Palestinian Children. Secretariat of the National Plan of Action for Palestinian Children, Save the Children. Ramallah.
- Ayer, Lynsay, Brinda Venkatesh, Robert Stewart, Daniel Mandel, Bradley Stein, and Michael Schoenbaum (2017). "Psychological Aspects of the Israeli-Palestinian Conflict: A Systematic Review." *Trauma, Violence, & Abuse*, 18, 322–338.
- Bertoni, Eleonora, Michele Di Maio, Vasco Molini, and Roberto Nisticò (2018). "Education is Forbidden: The Effect of the Boko Haram Conflict on Education in North-East Nigeria." *Journal of Development Economics*, forthcoming.
- Blattman, Christopher and Jeannie Annan (2010). "The Consequences of Child Soldiering." *Review of Economics and Statistics*, 92, 882–898.
- Blattman, Christopher and Edward Miguel (2010). "Civil War." *Journal of Economic Literature*, 48, 3–57.
- Brück, Tilman, Patricia Justino, Philip Verwimp, Alexandra Avdeenko, and Andrew Tedesco (2016). "Measuring Violent Conflict in Micro-Level Surveys: Current Practices and Methodological Challenges." World Bank Research Observer, 31, 29–58.
- B'Tselem (2007). *Human Rights in the Occupied Territories*. Annual Report. B'Tselem: The Israeli Information Center for Human Rights in the Occupied Territories, Jerusalem.
- Buvinić, Mayra, Monica Das Gupta, and Olga Shemyakina (2014). "Armed Conflict, Gender and Schooling." World Bank Economic Review, 28, 311–319.
- Calì, Massimiliano and Sami Miaari (2018). "The Labor Market Impact of Mobility Restrictions: Evidence from the West Bank." *Labour Economics*, 51, 136–151.
- Card, David (1999). "The Causal Effect of Education on Earnings." In *Handbook of Labor Economics*, Vol. 3A, edited by O. Ashenfelter and D. Card. Elsevier, Amsterdam pp. 1801–1863.
- Chamarbagwala, Rubiana and Hilcías E. Morán (2010). "The Human Capital Consequences of Civil War: Evidence from Guatemala." *Journal of Development Economics*, 94, 41–61.

CIA (2008). The World Factbook 2007. Central Intelligence Agency, Washington, DC.

- Cleveland, Benjamin and Kenn Fisher (2014). "The Evaluation of Physical Learning Environments: A Critical Review of the Literature." *Learning Environments Research*, 17, 1–28.
- de Groot, Olaf and Idil Goksel (2011). "Conflict and Education Demand in the Basque Region." Journal of Conflict Resolution, 55, 652–677.
- Di Maio, Michele and Tushar Nandi (2013). "The Effect of the Israeli-Palestinian Conflict on Child Labour and School Attendance in the West Bank." *Journal of Development Economics*, 100, 107–116.
- Di Maio, Michele and Roberto Nisticò (2018). "Effect of Parental Job Loss on Child School Dropout: Evidence from the Occupied Palestinian Territories." CSEF Working Paper 456, Centre for Studies in Economics and Finance (CSEF), University of Naples Federico II, Naples.
- Do, Quy-Toan and Lakshmi Iyer (2012). "Mental Health in the Aftermath of Conflict." In Oxford Handbook of the Economics of Peace and Conflict, edited by M. R. Garfinkel and S. Skaperdas. Oxford University Press.
- Dubow, Eric F., Paul Boxer, L. Rowell Huesmann, Khalil Shikaki, Simha Landau, Shira Dvir Gvirsman, and Jeremy Ginges (2009). "Exposure to Conflict and Violence across Contexts: Relations to Adjustment among Palestinian." *Children, Journal of Clinical Child & Adolescent Psychology*, 39, 103–116.
- Etkes, Haggay and Assaf Zimring (2015). "When Trade Stops: Lessons from the Gaza Blockade 2007–2010." *Journal of International Economics*, 95, 16–27.
- European Commission Education, Audiovisual and Culture Executive Agency (EACEA) (2012). *Higher Education in the Occupied Palestinian Territories*. European Union, Brussels.
- European Training Foundation (ETF) (2014). *Labour Market and Employment Policies in Palestine*. European Training Foundation Communication Department, Turin.
- European Union (2006). *Report of Election Observation Mission to West Bank and Gaza 2006*. European Union, Brussels.
- Gershenson, Seth and Erdal Tekin (2018). "The Effect of Community Traumatic Events on Student Achievement: Evidence from the Beltway Sniper Attacks." *Education Finance and Policy*, 13, 513–544.
- Giacaman, Rita, Anita Abdullah, Rula Abu Safieh, and Luna Shamieh (2002). Schooling at Gunpoint: Palestinian Children's Learning Environment in War like Conditions. The Ramallah/ al-Bireh/Beitunia Urban Center.
- Giacaman, Rita, Harry S. Shannon, Hana Saab, Neil Arya, and W. Thomas Boyce (2007). "Individual and Collective Exposure to Political Violence: Palestinian Adolescents Coping with Conflict." *European Journal of Public Health*, 17, 361–368.
- Glewwe, Paul and Michael Kremer (2006). "Schools, Teachers, and Education Outcomes in Developing Countries." In *Handbook of the Economics of Education*, Vol. 2, edited by E. Hanushek and F. Welch. Elsevier, Amsterdam, pp. 945–1017.
- Grogger, Jeffrey (1997). "Local Violence and Educational Attainment." *Journal of Human Resources*, 32, 659–682.
- Gvirsman, Shira Dvir, L. Rowell Huesmann, Eric F. Dubow, Simha F. Landau, Khalil Shikaki, and Paul Boxer (2014). "The Effects of Mediated Exposure to Ethnic-Political Violence on Middle East Youth's Subsequent Post-Traumatic Stress Symptoms and Aggressive Behavior." *Communication Research*, 41, 961–990.
- Hanushek, Eric A. and Steven G. Rivkin (2010). "Generalizations about Using Value-Added Measures of Teacher Quality." *American Economic Review*, 100(2), 267–271.
- Hanushek, Eric A. and Ludger Woessmann (2011). "The Economics of International Differences in Educational Achievement." In *Handbook of the Economics of Education*, Vol. 3, edited by E. Hanushek, S. Machin, and L. Woessmann. Elsevier, Amsterdam, pp. 89–200.
- Higgins, Steven Edward, Elaine Hall, Kate Wall, Pam Woolner, and Caroline McCaughey (2005). *The Impact of School Environments: A Literature Review*. The Design Council, London.
- Jaeger, David A. and M. Daniele Paserman (2008). "The Cycle of Violence? An Empirical Analysis of Fatalities in the Palestinian-Israeli Conflict." *American Economic Review*, 98(3), 1591– 1604.

- Jaeger, David A., Esteban F. Klor, Sami H. Miaari, and M. Daniele Paserman (2012). "The Struggle for Palestinian Hearts and Minds: Violence and Public Opinion in the Second Intifada." *Journal* of Public Economics, 96, 354–368.
- Justino, Patricia (2012). "Violent Conflict and Human Capital Accumulation." In *Elgar Handbook* of Civil War and Fragile States, edited by G. K. Brown and A. Langer. Edward Elgar Publisher, Cheltenham, UK.
- Justino, Patricia, Leone Marinella, and Paola Salardi (2014). "Short- and Long-Term Impact of Violence on Education: The Case of Timor Leste." World Bank Economic Review, 28, 320– 353.
- Kibris, Arzu (2015). "The Conflict Trap Revisited: Civil Conflict and Educational Achievement." Journal of Conflict Resolution, 59, 645–670.
- Koenker, Roger and Kevin F. Hallock (2001). "Quantile Regression: An Introduction." Journal of Economic Perspectives, 15(4), 143–156.
- Leon, Gianmarco (2012). "Civil Conflict and Human Capital Accumulation. The Long Term Effects of Political Violence in Peru." *Journal of Human Resources*, 47, 991–1022.
- Mansour, Hani and Daniel I. Rees (2012). "The Effect of Prenatal Stress on Birth Weight: Evidence from the Al-Aqsa Intifada." *Journal of Development Economics*, 99, 190–199.
- Mataria, Awad, Rita Giacaman, Angelo Stefanini, Nirmala Naidoo, Paul Kowal, and Somnath Chatterji (2009). "The Quality of Life of Palestinians Living in Chronic Conflict: Assessment and Determinants." *European Journal of Health Economics*, 10, 93–101.
- McGill, Tia M., Shannon R. Self-Brown, Betty S. Lai, Melissa Cowart-Osborne, Ashwini Tiwari, Monique LeBlanc, and Mary Lou Kelley (2014). "Effects of Exposure to Community Violence and Family Violence on School Functioning Problems among Urban Youth: The Potential Mediating Role of Posttraumatic Stress Symptoms." *Frontiers in Public Health*, 2, 1–8.
- McMullen, Steven C. and Kathryn E. Rouseb (2012). "School Crowding, Year-Round Schooling, and Mobile Classroom Use: Evidence from North Carolina." *Economics of Education Review*, 31, 812–823.
- Miaari, Sami (2009). "The Public-Private Wage Differential in the West Bank and Gaza Strip Before and During the Second Intifada." EUI Working Papers, MWP 2009/13, Florence.
- Miaari, Sami (2018). "An Analysis of the Public Private Wage Differential in the Palestinian Labour Market." *Defence and Peace Economics*, doi: 10.1080/10242694.2018.1473137.
- Michaelsen, Maren M. and Paola Salardi (2018). "Violence, Psychological Stress and Educational Performance during the "War on Drugs" in Mexico." Working Paper 595, University of Toronto, Toronto.
- Monteiro, Joana and Rudi Rocha (2017). "Drug Battles and School Achievement: Evidence from Rio de Janeiro's Favelas." *Review of Economics and Statistics*, 99, 213–228.
- Nicolai, Susan (2007). Fragmented Foundations: Education and Chronic Crisis in the Occupied Palestinian Territories. Lessons from the First Decade of a New Palestinian Education System, 1994–2005. UNESCO, Paris and Save the Children UK, London.
- Palestinian Central Bureau of Statistics (PCBS) (various years). Palestinian Labour Force Survey. Ramallah.
- Palestinian Central Bureau of Statistics (PCBS) (1997). Population Census. Ramallah.
- Palestinian Central Bureau of Statistics (PCBS) (2004). *Psychosocial Health Survey for Children*. Ramallah.
- Palestinian Central Bureau of Statistics (PCBS) (2009). *Internal Migration and Returnees in the Palestinian Territory*. Ramallah.
- Palestinian Ministry of Education and Higher Education (MOEHE) (2005). Assessment "15"—The Effect of the Israeli Occupation on the Palestinian Education 28/9/2000–9/7/2006. Ramallah.
- Pedersen, Jon, Mona Christophersen, Lena C. Endresen, and Pål Sletten (2001). *Paying a Price. Coping with Closures in Two Palestinian Villages.* FAFO Report 365. Oslo.
- Pynoos, Robert S., Calvin Frederick, Kathi Nader, William Arroyo, Alan Steinberg, Spencer Eth, Francisco Nunez, and Lynn Fairbanks (1987). "Life Threat and Posttraumatic Stress in School-Age Children." Archives of General Psychiatry, 44, 1057–1063.
- RAND (2015). The Cost of the Israeli-Palestinian Conflict. RAND Corporation, USA.

- Rivera-Batiz, Francisco L. and Lilian Marti (1995). A School System at Risk: A Study of the Consequences of Overcrowding in New York City Public Schools. Institute for Urban and Minority Education, Teachers College, Columbia University, New York.
- Rivkin, Steven G., Eric A. Hanushek, and John F. Kain (2005). "Teachers, Schools, and Academic Achievement." *Econometrica*, 73, 417–458.
- Rodriguez, Catherine and Fabio Sanchez (2012). "Armed Conflict Exposure, Human Capital Investments and Child Labor: Evidence from Colombia." *Defence and Peace Economics*, 23, 161–184.
- Roeser, Robert, Jacquelynne S. Eccles, and Karen R. Strobel (1998). "Linking the Study of Schooling and Mental Health: Selected Issues and Empirical Illustrations at the Level of the Individual." *Educational Psychologist*, 33, 153–176.
- Save the Children (2003). *Growing Up during Curfew. Safeguarding the Basic Rights of Palestinian Children.* Save the Children UK, London.
- Save the Children (2004). *Living behind Barriers: Palestinian Children Speak Up*. Save the Children UK, London.
- Sharkey, Patrick (2010). "The Acute Effect of Local Homicides on Children's Cognitive Performance." *Proceedings of the National Academy of Sciences, USA*, 107, 11733–11738.
- Shemyakina, Olga (2011). "The Effect of Armed Conflict on Accumulation of Schooling: Results from Tajikistan." *Journal of Development Economics*, 95, 186–200.
- Srour, Roney W. (2005). "Children Living under a Multi Traumatic Environment." Israel Journal of Psychiatry, 42, 88–89.
- Sultana, Ronald G. (2003). "An EMIS for Palestine. The Education Management Information System in the West Bank and Gaza Strip." *Mediterranean Journal of Educational Studies*, 7, 61–92.
- Sultana, Ronald G. (2006). "Education in Conflict Situations: Palestinian Children and Distance Education in Hebron." *Mediterranean Journal of Educational Studies*, 11, 49–81.
- Swee, Eik (2015). "On War and Schooling Attainment: The Case of Bosnia and Herzegovina." European Journal of Political Economy, 40, 158–172.
- Tansel, Aysit and Yousef Daoud (2011). "Comparative Essay on Returns to Education in Palestine and Turkey." IZA DP No. 5907.
- Thabet, Abdel Aziz Mousa, Yehia Abed, and Panos Vostanis (2002). "Emotional Problems in Palestinian Children Living in a War Zone: A Cross-sectional Study." *Lancet*, 359, 1801–1805.
- Thabet, Abdel Aziz Mousa, Ahmed Abu Tawahina, Eyad El Sarraj, and Panos Vostanis (2008). "Exposure to War Trauma and PTSD among Parents and Children in the Gaza Strip." *European Child & Adolescent Psychiatry*, 17, 191–199.
- UNDP (2015). Development for Empowerment. The 2014 Palestine Human Development Report. UNDP, Genève.
- UNESCO (2007). Global Education Digest. Institute for Statistics. Data Centre. UNESCO, Paris.
- UNESCO (2011a). The Hidden Crisis: Armed Conflict and Education. EFA Global Monitoring Report. UNESCO, Paris.
- UNESCO (2011b) World Data on Education: Palestine. UNESCO, Paris.
- UNICEF (2010). The Situation of Palestinian Children in the Occupied Palestinian Territory, Jordan, Syria and Lebanon. UNICEF, Paris.
- UN-OCHA-OPT (various issues). The Monthly Humanitarian Monitor. East Jerusalem.
- UNRWA (2003). *Emergency Appeal June–December 2003*. Department of External Relations, UNRWA Headquarters, Gaza, pp. 3–6.
- Valente, Christine (2013). "Education and Civil Conflict in Nepal." *World Bank Economic Review*, 28, 254–383.
- Verwimp, Philip, Patricia Justino, and Tilman Brück (forthcoming). "The Microeconomics of Violent Conflict." *Journal of Development Economics*, in press. https://doi.org/ 10.1016/j.jdeveco.2018.10.005.
- Wilson, Cody and Beth Spenciner Rosenthal (2003). "The Relationship between Exposure to Community Violence and Psychological Distress among Adolescents: A Meta-analysis." *Violence* and Victims, 18, 335–352.

World Bank (2002). *West Bank and Gaza: An Evaluation of Bank Assistance*. Operations Evaluation Department. World Bank, Washington, DC.

World Bank (2006). West Bank and Gaza: Education Sector Analysis. World Bank, Washington, DC.

World Bank (2007a). Movement and Access Restrictions in the West Bank: Uncertainty and Inefficiency in the Palestinian Economy. World Bank, Washington, DC.

- World Bank (2007b). West Bank and Gaza Public Expenditure Review: From Crisis to Greater Fiscal Independence. World Bank, Washington, DC.
- World Bank (2008). *The Economic Effects of Restricted Access to Land in the West Bank*. Social and Economic Development Group Finance and Private Sector Development Middle East and North Africa Region. World Bank, Washington, DC.
- World Bank (2011). West Bank and Gaza Coping with Conflict? Poverty and Inclusion in the West Bank and Gaza. World Bank, Washington, DC.

Supplementary Data

Supplementary data are available at *JEEA* online.